Author's response to reviews

Title: Role of Dectin-1 in the innate immune response of mice corneal epithelial cells to Aspergillus fumigatus

Authors:

qiang xu (yankeuxiqiang@126.com)
guiqiu zhao (zhaoquigiu_good@126.com)
jing lin (yankejina@126.com)
qian wang (yankeeswqian@126.com)
liting hu (yankehu@126.com)
zhao jiang (yankejiangzhao@126.com)

Version: 5
Date: 2 April 2015

Author's response to reviews: see over
Dear editors,
All email address of each co-authors were added to the Title page.
Acknowledgment section was added to the manuscript
A separate page for Title Page was provided.
Thank you very much!